Author's response to reviews

Title: Mental health priorities in Vietnam: A mixed-methods analysis

Authors:

Maria E Niemi (maria.niemi@ki.se)
Huong T Thanh (huongtran2008@gmail.com)
Tran Tuan (trantuanrtccd@gmail.com)
Torkel Falkenberg (torkel.falkenberg@ki.se)

Version: 2 Date: 29 April 2010

Author's response to reviews:

Dear editor,

I submit herewith the revisions requested. These include a new version of table 2 (page 26), where we have removed the affiliations of the interviewees in order to protect their anonymity. We have added an annotation at the bottom of the table, noting that interviews were generally conducted with leading representatives of the institutions/organisations, as we deem this necessary information in order to assure validity of the findings. Also as requested, I attach the interview guide that was used for data collection.

Best regards, Maria Niemi